Author's response to reviews

Title: A novel biomarker TERTmRNA is applicable for early detection of hepatoma

Authors:

Norimasa Miura (mnmiura@hotmail.com)
Yukio Osaki (yosk@osaka-med.jrc.or.jp)
Miki Nagashima (miki.nagashima@celery.ocn.ne.jp)
Michimori Kohno (kohno@web-sanin.co.jp)
Kensho Yorozu (incepk@yahoo.co.jp)
Kohei Shomori (shomori@med.tottori-u.ac.jp)
Takamasa Kanbe (tikanbe@sanmedia.or.jp)
Kenji Oyama (oyamaknj@hotmail.com)
Yukihiro Kishimoto (kisimoto075@sanin-rosai-h.yonago.tottori.jp)
Shigeo Maruyama (maruyama-clinic@athena.ocn.ne.jp)
Eijiro Noma (syokaki@fukuoka-u.ac.jp)
Yutaka Horie (y-horie@saiseikai-gotsu.jp)
Masatoshi Kudo (m-kudo@med.kindai.ac.jp)
Seigo Sakaguchi (syokaki@fukuoka-u.ac.jp)
Yasuaki Hirooka (yhirooka@med.tottori-u.ac.jp)
Hisao Ito (ito@med.tottori-u.ac.jp)
Hironaka Kawasaki (kawasaki002@sanin-rosai-h.yonago.tottori.jp)
Junichi Hasegawa (hasegawa@med.tottori-u.ac.jp)
Goshi Shiota (gshiota@med.tottori-u.ac.jp)

Version: 8 Date: 12 April 2010

Author's response to reviews:

I corrected the author's contributions as you pointed out. I am terribly sorry for bothering you.